Author's response to reviews

Title: Early experience with Single-Incision Laparoscopic Surgery (SILS) for the placement of a gastrostomy in a 10-year-old girl: a case report

Authors:

Kim Vanderlinden (kim.vanderlinden@uzbrussel.be)
Nele Van De Winkel (nele.vandewinkel@uzbrussel.be)
Antoine De Backer (antoine.debacker@uzbrussel.be)
Georges Delvaux (georges.delvaux@uzbrussel.be)
Kristel De Vogelaere (kristel.devogelaere@uzbrussel.be)

Version: 4 Date: 9 August 2012

Author's response to reviews: see over
Dear Editor--in-Chief,

Thank you for sending us the thorough review and for allowing us to submit a revised paper addressing the concerns of both Referees.

Our resubmission includes 2 files:
a Word version of our revised Manuscript with the revised parts of the text marked in green print;
a Word version of our responses to the Associate Editor and Referees with restatement of their comments (italic print), followed by our responses.

As we were able to address the issues raised, we sincerely hope that you will be able to accept our Revised Manuscript in its current form.

The present manuscript has not been reported elsewhere. Submission for publication is fully approved by all authors.

Hoping you will take this manuscript into consideration for publication we remain,
Yours sincerely,

Dr Kim Vanderlinden
University Hospital Brussel
Jette
Belgium

kim.vanderlinden@uzbrussel.be